Author's response to reviews

Title: Study protocol for a two-armed cluster-randomised trial to assess whether indoor residual spraying can provide additional protection against clinical malaria over current best practice of long-lasting insecticidal mosquito nets in The Gambia

Authors:

Margaret Pinder (mpinder@mrc.gm)
Musa Jawara (mjawara@mrc.gm)
Lamin BS Jarju (lbsjarju@yahoo.co.uk)
Balla Kandeh (ballakandeh@yahoo.co.uk)
David Jeffries (djeffries@psi.org)
Manuel F Lluberas (lluberas@hdhudson.com)
David J Conway (david.conway@lshtm.ac.uk)
Steve W Lindsay (steve.lindsay@lshtm.ac.uk)

Version: 2 Date: 10 May 2011

Author's response to reviews:

Dear Andrew

I have now uploaded my manuscript without the figures. The two figures are provided as separate files.

Sorry for causing a problem.

best wishes

Steve Lindsay